Author's response to reviews

Title: Part-time and full-time medical specialists, are there differences in allocation of time?

Authors:

Judith D. de Jong (j.dejong@nivel.nl)
Phil Heiligers (p.heiligers@nivel.nl)
Peter P. Groenewegen (p.groenewegen@nivel.nl)
Lammert Hingstman (l.hingstman@nivel.nl)

Version: 4 Date: 17 February 2006

Author’s response to reviews:

Formatting changes are made as requested:
*Title and abstract are made identical in the manuscript and submission system.
*The list of authors in the manuscript is written exactly as they are in the submission system.
*Box 1 is renamed Table 1 and the other tables are renumbered. References to the box and tables are amended to reflect these changes.